Title: Expression of centromere protein F (CENP-F) associated with higher FDG uptake on PET/CT, detected by cDNA microarray, predicts high-risk patients with primary breast cancer

Authors:

Shigeto Ueda (syueda2000@yahoo.co.jp)
Nobuo Kondoh (NouoK@ndmc.ac.jp)
Hitoshi Tsuda (htsuda@ndmc.ac.jp)
Souhei Yamamoto (SouhYM@ndmc.ac.jp)
Junji Yamamoto (JunYM@ndmc.ac.jp)
Katsumi Tamura (KatsumiTM@tokopet.co.jp)
Jiro Ishida (JiroIS@tokopet.co.jp)
Yoshiyuki Abe (YoshiAB@tokopet.co.jp)
Mikio Yamamoto (mikioYM@ndmc.ac.jp)
Hidetaka Mochizuki (mochidukiH@ndmc.ac.jp)

Version: 2 Date: 18 July 2008

Author's response to reviews:

July 18, 2008
Editor-in-Chief Melissa Norton, MD

Dear Dr. Melissa Norton,

Title: Expression of centromere protein F (CENP-F) associated with higher 18F-FDG uptake on PET/CT, detected by cDNA microarray, predicts high-risk patients with primary breast cancer

Enclosed is my manuscript for you to consider for possible publication in BMC cancer. This study categorizes full papers report. No portion of the paper has been published, and the manuscript is not under consideration of any other journal. All authors have read this report and agreed to submission to the journal. There are 3 figure, 3 tables and totally 5622 words in this report.

Thank you for considering the paper for publication in BMC cancer.

Sincerely,

Hitoshi Tsuda, MD,PhD
Departments of Basic Pathology, National Defense Medical College
Namiki 3-2, Tokorozawa, Saitama, Japan
e-mail: htsuda@ndmc.as.jp
TEL. 81-4-2995-1633
FAX. 81-4-2991-1613